(As amended November 4, 2011 by Addendum No. 1)

ALBERTA INFRASTRUCTURE REQUEST FOR QUALIFICATIONS

FOR THE

DESIGN, BUILD, FINANCE, AND MAINTENANCE

OF

THIRTEEN NEW SCHOOLS IN CENTRAL AND SOUTHERN ALBERTA, CANADA

Response Deadline: Wednesday, November 16, 2011 before 2:00 p.m. (Edmonton time)

Government of Alberta ■ Infrastructure

Date of Issuance: October 14, 2011

TABLE OF CONTENTS

1	INT	RODUCTION	1
	1.1	Summary of the Business Opportunity	1
	1.2	Intentionally Deleted	3
	1.3	Objectives of this RFQ	3
	1.4	Instructions to Interested Parties	5
	1.5	Format of Response	
	1.6	Addenda to this RFQ	6
	1.7	Communications	6
	1.8	No Lobbying	
	1.9	Intentionally Deleted	
	1.10	Available Information	
	1.11	No Liability	
	1.12	Defined Terms	
2		E ASAP III DBFM PROJECT	
	2.1	Description of the ASAP III DBFM Project	
	2.2	Core School Concept	
	2.3	Design and Construction	
	2.4	Maintenance and Renewal	
	2.5	Financing and Payment	
	2.6	Procurement Process	
	2.7	Intentionally Deleted	
	2.8	DBFM Agreement and Risk Allocation	
-	2.9	ASAP III DBFM Project Schedule	
3		REQUIREMENTS	
	3.1	Ineligibility	
	3.2	Team Name and Lead Contact	
	3.3	Confidentiality of Responses and the Collection of Personal Information	16
	3.4	Publicity	
	3.5	No Collusion	
	3.6	No Common Ownership	
	3.7	Non-Conforming Responses.	
1	3.8	Expenses	18 19
4	4.1	Team Members	
	4.1 4.2	Key Individuals	
	4.2 4.3	Substitutions	
5		JR EXPERIENCE	
5	5.1	Team Members' Project Experience	
	5.2	Key Individuals' Experience	
6	• • =	JR APPROACH	
0	6.1	Project Lead	
	6.2	Design-Construction Team	
	6.3	M&R Team	
	6.4	Financing Team	
	0.7		20

7	YOU	JR ADVANTAGE	27
		Why We Should Be Selected to Participate in the RFP Process	
		EVALUATION	
	8.1	Completeness and Compliance Check	28
	8.2	Scoring Matrix	28
	8.3	Risk Assessment of Respondents	29
	8.4	Basis of Evaluation	29
	8.5	Clarifications and Interviews	29
	8.6	Debriefing	29

Appendix "A" (article 1.12 - Defined Terms) Appendix "B" (articles 3.2, 4.1, 4.2/5.2 and 5.1 - Tables)

Table 3.2: Contact Organization and Contact Individual Table 4.1A: Team Members

Table 4.1B: Project Lead Team Member Name and Role

Table 4.1C: Project Lead Team Member Legal Status

Table 4.1D: Design-Construction Team Members Name and Role

Table 4.1E: Design-Construction Team Member Legal Status

Table 4.1F: M&R Team Name and Role

Table 4.1G: M&R Team Member Legal Status

Table 4.1H: Financing Team Member Name and Role

Table 4.11: Financing Team Member Legal Status

Table 4.2 / 5.2A: Project Lead Team Key Individuals

Table 4.2 / 5.2B: Design-Construction Team Key Individuals

Table 4.2 / 5.2C: M&R Team Key Individuals

Table 4.2 / 5.2D: Financing Team Key Individuals

Table 5.1A: Project Lead Team Member Experience

Table 5.1B: Design-Construction Team Member Experience

Table 5.1C: M&R Team Member Experience

Table 5.1D: Financing Team Member P3 or Relevant Project Experience

Table 7.1: Your Advantage - Collective Experience

Exhibit "A" (articles 1.1 and 2.1 – school sites)

1 INTRODUCTION

1.1 Summary of the Business Opportunity

Alberta Infrastructure (**"INFRA**") has received Government of Alberta approval to proceed with the procurement of the design, build, finance and Maintenance ("**DBFM**") of thirteen new schools in central and southern Alberta (the "**ASAP III DBFM Project**").

The ASAP III DBFM Project is a top priority for INFRA and will build upon the successes enjoyed in the prior two Alberta Schools Alternative Procurements that closed on September 12, 2008 and April 15, 2010 ("**ASAP I Project**" and "**ASAP II Project**", respectively). As for those prior projects, INFRA has formed the ASAP III DBFM Project team, which includes representatives from INFRA, Alberta Education, Alberta Justice, Ministry of Treasury Board and Enterprise and Alberta Finance, to manage the procurement of the ASAP III DBFM Project on behalf of the Government of Alberta.

The 13 schools to be built as part of the ASAP III DBFM Project are located in Airdrie, Beaumont, Brooks, Chestermere, Cochrane, Medicine Hat, Penhold, and Red Deer. These 13 new schools will open for school board use in 2014.

The new schools will be built to provincial standards and will be based upon core school designs. The permanent core building with the essential elements of a school will be complemented by high performance modular classrooms that can be added or removed as required. The schools will be built to Leadership in Energy and Environmental Design (LEEDTM) Silver standards.

The proposed capital investment in the ASAP III DBFM Project schools supports the Government of Alberta's long-term vision for lifelong learning and managing growth pressures. It will also support the Business Plans of Alberta Education and Alberta Infrastructure.

In addition to the design and construction, the ASAP III DBFM Project will also include Maintenance and Renewal ("**M&R**") over a 30-year period and a financing component. Ownership of the ASAP III DBFM Project schools will remain with the school boards and/or the municipalities, as applicable.

It is anticipated that at some school sites there will be "community partnership" space designed and built as part of the ASAP III DBFM Project. These spaces will be used by the community or leased to third party entities. The Successful Proponent will be responsible for the design and construction of these spaces but the Successful Proponent will not be responsible for any associated maintenance or renewal. The community partnership space may be attached to the school core structure or may be freestanding. Where attached to a school's core structure, the core school designs will provide for the separation of these spaces with fire walls from the core school structure. The core school designs will also provide for separate mechanical and electrical systems to facilitate the Successful Proponent not having any long-term responsibility for these spaces.

The following will remain the responsibilities of the school boards:

• Education services for students;

- Custodial services including day-to-day monitoring of boilers, and scheduling of heating and ventilation; and
- Supply, installation, maintenance and replacement of furniture and equipment.

Access to and use of the new schools will be managed by the school boards in the same manner as existing schools.

The specific school programs and sites were determined in consultation with:

- (a) Black Gold Regional Division No. 18;
- (b) Chinook's Edge School Division No. 73;
- (c) Grasslands Regional Division No. 6;
- (d) Greater North Central Francophone Education Region No. 2 (Conseil scolaire Centre-Nord);
- (e) Greater Southern Public Francophone Education Region No. 4 (Conseil scolaire du Sud de l'Alberta);
- (f) Greater Southern Separate Catholic Francophone Education Region No. 4 (Conseil scolaire catholique et francophone du Sud de l'Alberta);
- (g) Medicine Hat School District No. 76;
- (h) Red Deer Catholic Regional Division No. 39;
- (i) Red Deer Public School District No. 104;
- (j) Rocky View School Division No. 41; and
- (k) St. Thomas Aquinas Roman Catholic Separate Regional Division No. 38,

(collectively the "ASAP III School Boards").

The locations of the ASAP III DBFM Project schools are set out in Exhibit "A", below. The ASAP III DBFM Project is described in more detail in article 2 of this Request for Qualifications ("**RFQ**").

The ASAP III DBFM Project will be procured as a DBFM. INFRA intends that the RFQ phase of the procurement will be followed by a Request for Proposals ("**RFP**") which will culminate in the signing of a DBFM Agreement, assuming competitive and affordable Proposals are received in response to the RFP. The procurement process and the DBFM Agreement will be based in large measure upon the successful processes and DBFM Agreement realized in the ASAP II Project.

The DBFM Agreement for the ASAP III DBFM Project will award to the Successful Proponent responsibility to design, construct and finance the ASAP III DBFM Project, and to maintain it for a term of 30 years following scheduled completion of the schools. Following completion of the schools, the Successful Proponent will receive monthly payments (as described in more detail in article 2.5, below) for the Term of the DBFM Agreement based on availability of the schools and compliance with the performance specifications.

The ASAP III DBFM Project is anticipated to yield a number of long term outcomes:

- Construction of schools in locations where students live;
- Support for Alberta Education's "Class Size Initiative";
- Cost certainty over the life cycle of the schools;

- Reduction in delivery timeframe and dedicated public sector resources necessary for the provision of schools;
- Effective 30-year 'warranty' period for school facilities;
- Equity of standards for school facilities within the province;
- High quality, LEEDTM certified buildings (Silver Standard); and
- Provision of an innovative, repeatable, transparent and accountable process for school facility project delivery.

INFRA anticipates providing partial funding for the ASAP III DBFM Project in an amount of approximately 50% of the Successful Proponent's capital cost of the ASAP III DBFM Project. The Government of Alberta is currently considering the appropriate amount of partial funding, the method of calculating the payment of such partial funding and the appropriate timing for payment of such partial funding. The final decisions on these points will be incorporated into the final draft of the DBFM Agreement issued to the Proponents prior to the RFP's SR Package 3 Submission deadline as set out in the Project Schedule in article 2.9, below. The Successful Proponent will supply financing for the balance of the capital costs of the ASAP III DBFM Project.

INFRA intends that the RFP will be issued and conducted in accordance with the schedule set out in the ASAP III DBFM Project Schedule in article 2.9, below.

The deadline for submitting a Response to this RFQ is set out in article 1.4, below.

1.2 *Intentionally Deleted*

1.3 Objectives of this RFQ

This RFQ is the first stage in the DBFM procurement process. The purpose of this RFQ is:

- to confirm sufficient interest among private sector participants to proceed effectively with the DBFM procurement process; and
- to identify three teams (each, a "**Consortium**") of qualified private sector participants for participation in the RFP phase of the DBFM procurement process.

The objectives of INFRA for the ASAP III DBFM Project and the DBFM procurement process are:

- to expeditiously complete the procurement process;
- to complete construction of the ASAP III DBFM Project so that all schools achieve "Availability" (as will be defined in the DBFM Agreement) in 2014;
- to achieve "value for money" over the life of the ASAP III DBFM Project, i.e., during the design-build phase and for a 30-year period thereafter (the "**Term**");

- to ensure that the ASAP III DBFM Project is designed, built and maintained in a manner that supports effective teaching and learning and allows, as much as possible, for the operation of the schools to be indistinguishable from schools procured in a traditional manner from the perspective of school users;
- to ensure that the ASAP III DBFM Project is designed, built and maintained in an environmentally sound manner and in a manner that ensures the health and safety of students and other school users; and
- to ensure that the school infrastructure meets specified quality requirements when it is "handed back" to the school boards at the end of the Term.

Accordingly, the principal objective of this RFQ is to identify, from among Consortia responding to this RFQ (the "**Respondents**"), the three Respondents ("**Proponents**") who INFRA considers are most qualified to further the above objectives for the ASAP III DBFM Project and to participate in the remainder of the ASAP III DBFM Project procurement process. Specifically, by this RFQ process INFRA intends to identify the three Respondents who, in INFRA's assessment, are most likely to:

- engage in the procurement process through to final bids;
- submit highly competitive final bids that will achieve value for money for the ASAP III DBFM Project;
- achieve financial close in a timely manner; and
- if selected as the Successful Proponent, design, build and maintain the schools effectively and efficiently, in a manner that is financially and environmentally sound, and appropriately protects the health and safety of students and other school users.

INFRA believes that assessment of the Respondents' respective ability to satisfy the above criteria and to meet the above objectives turns on four key questions:

- Who are you? -- Who are the key business entities ("Team Members") on your team, and who are the individuals ("Key Individuals") that will play a lead role on behalf of those Team Members? Article 4 of this RFQ addresses this key question.
- What have you done? -- What pertinent experience, knowledge and skills do your Team Members and their Key Individuals have? Article 5 of this RFQ addresses this key question.
- What is your approach? -- How do you plan to structure your Consortium and your approach to each of the project lead role, design-build, the Maintenance and Renewal, and the financing of the ASAP III DBFM Project? Article 6 of this RFQ addresses this key question.

• What is your advantage? -- What are your significant advantages, whether of experience, organization, methodology or innovation, that suggest a relatively high probability that your Consortium will be able to deliver the winning bid and thereafter successfully carry out the ASAP III DBFM Project through to the end of the Term? Article 7 of this RFQ addresses this key question.

1.4 Instructions to Interested Parties

Registration

All interested parties requesting access to the SharePoint server site (see article 1.10) are required to register with INFRA ("**Registered Parties**") through:

Bouvana Sai Alternative Procurement Office Main Floor, Infrastructure Building 6950 – 113 Street Edmonton, Alberta, Canada T6H 5V7 Email: <u>Bouvana.sai@gov.ab.ca</u> Phone: (780) 644-8400 Fax: (780) 644-6402

Only Registered Parties will be entitled to have access to the electronic information SharePoint Server site. Once an interested party has made a request to access the SharePoint server site it may take up to 2 Business Days for that party to gain access.

Submission of Responses

Responses to this RFQ must:

- be in the format described in article 1.5 of this RFQ;
- clearly identify the Response package as relating to "Alberta Schools Alternative Procurement III 2011, Request for Qualifications"; and
- be addressed to the attention of and physically delivered to the address set out below before 2:00 p.m. Edmonton time on **Wednesday**, **November 16, 2011.**

Attention:	Office of the Tender Administrator
Location:	Alberta Infrastructure
	2 nd Floor, Infrastructure Building
	6950 – 113 Street
	Edmonton, Alberta, Canada
	T6H 5V7

Responses submitted by fax or e-mail are not acceptable and will not be considered or evaluated.

All Responses become the property of INFRA upon their submission and will not be returned.

1.5 Format of Response

Your Response must:

- be entirely in the English language;
- consist of all of the information required by this RFQ including, but not limited to, those requirements set out in article 3.2 and articles 4, 5, 6 and 7;
- include one electronic copy (in PDF format (searchable PDF, if possible) with permission to copy and print) and six original hard copies printed from the electronic copy so as to reduce the likelihood of conflict between the forms of the Response; and
- in furtherance of confidentiality, include in a **SEPARATE SEALED ENVELOPE MARKED "CONFIDENTIAL FINANCIAL INFORMATION"** one original hard copy of all financial statements, annual reports, bank references and alternative information indicated in article 4.1 that are part of the Response.

In the event of a discrepancy between the electronic and hard copy forms of any Respondent's Response, INFRA will exercise its discretion to determine which form will be used for evaluation purposes.

1.6 Addenda to this RFQ

INFRA may make modifications to this RFQ, including but not limited to extending the RFQ closing time, by issuing addenda. Hard copy recipients of the RFQ from the 'Office of the Tender Administrator' as set out in article 1.4, above, will receive RFQ addenda via fax or courier. Prospective Respondents who access the RFQ via the Alberta Purchasing Connection ("**APC**") website will receive an e-mail notification of any addenda posted to APC. Such addenda will amend and form part of the RFQ. It is the responsibility of the Respondent to ensure that the Respondent's Response is in accordance with all addenda to the RFQ.

1.7 Communications

All correspondence and contact by Registered Parties or Respondents (including their Team Members and Key Individuals) with INFRA in relation to this RFQ must be directly and only with the following individual (the "**Designated Representative**") designated by INFRA:

Fred DeMott Project Director - ASAP III Alternative Procurement Office Alberta Infrastructure Main Floor, Infrastructure Building 6950 – 113 Street Edmonton, Alberta, Canada T6H 5V7 E-mail: fred.demott@gov.ab.ca Phone: (780) 644-8401 Fax: (780) 644-8402

Registered Parties who wish to ask questions regarding this RFQ must submit them by a letter to the Designated Representative (which may be by e-mail attachment in PDF form).

INFRA may decline to respond to questions received less than 7 Business Days before the RFQ closing date indicated in article 1.4, above.

INFRA will, where in its assessment confidentiality is not an issue and its response to a particular question is likely to be of interest to other Registered Parties, distribute to all Registered Parties copies of questions and INFRA's responses or a written addendum to the RFQ to ensure that all Registered Parties base their Responses, should they submit Responses, on the same information.

1.8 No Lobbying

Interested parties, Registered Parties, Respondents and their Team Members and Key Individuals are strictly prohibited from engaging in any form of political or other lobbying whatsoever in relation to the ASAP III DBFM Project or with a view to influencing the outcome of this RFQ process. Failure to comply with this provision may result in disqualification of that Consortium from the RFQ process or, if INFRA becomes aware of a breach of this article after the RFP has been issued, from the RFP process.

1.9 Intentionally Deleted

1.10 Available Information

All Registered Parties will be given access to a SharePoint Server site after registration with INFRA, which site will contain the Tables 3.2, 4.1, 4.2/5.2, 5.1 and 7.1 in electronic Word format.

Additional information may be added to the SharePoint Server site if or when available, including site data sheets for each of the 13 school sites.

The ASAP II Project DBFM Agreement and Schedules are located on Alberta Education's website at: <u>http://education.alberta.ca/department/ipr/p3project/asapii/dbfm.aspx</u>

INFRA's department website is located at <u>http://www.infrastructure.alberta.ca</u>. This website contains information on the general policies related to school buildings including design, construction and maintenance. The webpage is located under the "buildings & lands: schools" menu at: <u>http://www.infrastructure.alberta.ca/516.htm</u>

INFRA also maintains a web-based "Technical Resource Centre-Buildings". The Technical Resource Centre is located under the "technical resources: technical resource centre – buildings: supported infrastructure" menu at: <u>http://www.infrastructure.alberta.ca/505.htm</u>

INFRA's design and construction standards and guidelines for schools is located at: <u>http://www.infrastructure.alberta.ca/Content/docType387/Production/designconstruction.pdf</u> This document may be accessed through both of the above noted webpages.

While the information accessible through the above noted websites has been provided in good faith, INFRA does not represent such information to be accurate, comprehensive or to have been independently verified. Neither INFRA nor any of its elected officials, officers, employees, agents, or advisors accepts any liability or responsibility for the adequacy, accuracy or completeness of, or make any representation or warranty, express or implied, with respect to the information accessible through the above noted websites. Any liability therefor is hereby expressly disclaimed.

1.11 No Liability

INFRA does not, by issuing this RFQ or by any communication or documentation made or provided in connection with this RFQ, incur any duty of care or contractual obligation to any interested party, Registered Party, Respondent, Team Member, Key Individual or other person, and expressly disclaims any liability or obligation to any interested party, Registered Party, Respondent, Team Member, Key Individual or other person in connection with this RFQ. Statements in this RFQ of INFRA's expectations in relation to the ASAP III DBFM Project, the RFQ Process, the ASAP III DBFM Project schedule and the RFP process are relied upon or acted upon by interested parties, Registered Parties, Respondents, Team Members, Key Individuals and other persons solely at their own risk.

INFRA may amend, suspend, postpone, cancel, or extend the closing of this RFQ or any future stage of the procurement of the ASAP III DBFM Project without incurring liability to any interested party, Registered Party, Respondent, Team Member, Key Individual or other persons.

1.12 Defined Terms

Defined terms used in this RFQ are defined as they are introduced and capitalized throughout. For convenience, a summary of defined terms is attached as Appendix "A".

2 THE ASAP III DBFM PROJECT

2.1 Description of the ASAP III DBFM Project

INFRA intends (but see the disclaimer in article 1.11, above) that the ASAP III DBFM Project will comprise the design and construction of the following thirteen new schools in southern and central Alberta. See Exhibit "A" to this RFQ for maps indicating the locations of the ASAP III DBFM Project schools described below.

ASAP III School Boards	Municipality	Approximate Address	Grade Structure	Core (Permanent Structure) Area - m2	Total Number of Modulars Delivered at School Availability	Approximate Student Capacity of School with Modular Classrooms Delivered at School Availability
Rocky View School Division No. 41	City of Airdrie	186 Sagewood Boulevard SW	6-8	6,462	16	900
Rocky View School Division No. 41	City of Airdrie	2654 Chinook Winds Drive SW	9-12	10,547	0	1,000
Greater Southern Public Francophone Education Region No. 4	City of Airdrie	1940 Reunion Boulevard NW	K-12	3,596	6	400
Black Gold Regional Division No. 18	Town of Beaumont	Dansereau Meadows (Phase 9, SDA-11- 01)	K-9	6,312	0	700
St. Thomas Aquinas Roman Catholic Separate Regional Division No. 38	Town of Beaumont	Dansereau Meadows (Phase 9, SDA-11- 01)	K-9	4,200	0	400
Grasslands Regional Division No. 6	City of Brooks	Upland Boulevard	K-6	2,798	12	450
Red Deer Public School District No. 104	City of Red Deer	300 Timothy Drive.	K-5	3,459	8	500
Greater North Central Francophone Education Region No. 2	City of Red Deer	Allan Street and Avery Street	K-12	2,810	2	250
Red Deer Catholic Regional Division No. 39	City of Red Deer	60 Clearview Drive	K-5	3,088	0	300
Rocky View School Division No. 41	Town of Chestermere	325 Kinniburgh Boulevard	K-9	5,964	0	500
Medicine Hat School District No. 76	City of Medicine Hat	751 Strachan Road SE	K-9	5,964	4	600
Chinook's Edge School Division No. 73	Town of Penhold	1 Waskasoo Avenue	7-12	4,519	6	500
Greater Southern Separate Catholic Francophone Education Region No. 4	Town of Cochrane	Quigley Drive and West Terrance Crescent	K-9	2,694	0	200
		Totals	5	62,413	54	6,700

2.2 Core School Concept

INFRA has mandated a "core school design" concept for all future Alberta schools. These designs provide consistent facilities that meet provincial school standards and guidelines and will support the Alberta curriculum. The designs can be easily adapted to different sites across the province and will allow for changing demographics. The designs take into account the requirements for attaining LEEDTM Silver Certification.

The ASAP III DBFM Project will use core school designs for K-5, K-6, K-9, K-12, 6-8, 7-12 and 9-12 schools.

The ASAP III DBFM Project schools will have a permanent core building with all the essential elements of a school, including a gymnasium, library, administration offices, washrooms, classrooms and special teaching spaces for subjects such as music, art, science, and career and technology studies. High performance modular classrooms will be added to and removed from the core building as needed. These designs allow schools to grow or to reduce in size, responding to changes in student enrolment.

High performance modular classrooms will provide improved durability and will retain their shape after multiple moves. The new modular classrooms are constructed and shipped as complete 24-foot-wide units and placed on screw pile foundations. This means the classrooms do not have joint lines that can shift and misalign, protecting them from unwanted air and moisture penetration. The modular classrooms look like permanent classrooms on the inside, with painted drywall, drop ceilings and venting windows. Overall, the specifications of modular classrooms are aimed at improving the health and comfort of students and teachers. The modular classroom output specification takes into account the requirements for attaining LEEDTM Silver Certification.

The table at article 2.1, above, sets out the anticipated number of modular classrooms initially required for the ASAP III DBFM Project.

2.3 Design and Construction

The Successful Proponent will design and construct the ASAP III DBFM Project schools in accordance with the Technical Requirements of the DBFM Agreement (as defined therein) and must comply with all applicable national, provincial and municipal laws, codes, standards, regulations and by-laws.

The DBFM Agreement will also include the supply and installation of the required modular classrooms. If required, the Successful Proponent shall add or remove modular classrooms and make good the building envelope and site upon notice from the Province during the Term of the DBFM Agreement. The DBFM Agreement will include provisions for payment to the DBFM contractor for this extra work, including if applicable, the provision of additional modular classrooms (see article 2.5, below).

INFRA will be making applications for development permits ("**DPs**") to the respective municipalities during the RFP process. However, the Successful Proponent will be responsible for any DP requirements that remain outstanding at the time of the execution of the DBFM Agreement.

2.4 Maintenance and Renewal

The Successful Proponent will be responsible for the Maintenance and Renewal ("**M&R**") of the ASAP III DBFM Project schools. The Successful Proponent will perform M&R in compliance with the Technical Requirements of the DBFM Agreement (as defined therein) and must comply with all applicable national, provincial and municipal laws, codes, standards, regulations and by-laws.

The school boards will be responsible for the supply and maintenance of the furniture and equipment required in the schools. They will also provide custodial services for each school. The scope of custodial services has been determined in consultation with the school boards and will consist of occupant support (those duties which support educational instruction activities), and cleaning (maintaining standard of cleanliness). Day-to-day monitoring of boilers and scheduling of heating and ventilation may be performed by the school boards. Following the Successful Proponent's initial installation of each school's information, communications and telecommunications ("**ICT**") cabling, the school boards will be responsible for the maintenance, repair and replacement of all ICT cabling during the remainder of the Term of the DBFM Agreement and for all school board supplied furniture and equipment including but not limited to desks, chairs, shelving, bookcases, whiteboards, computers, books, supplies, and security equipment.

School board responsibilities for custodial services are anticipated to be largely the same as for the prior ASAP II Project (see article 1.10 for more information).

2.5 Financing and Payment

Subject to the funding provided by INFRA described below, the Successful Proponent will supply all necessary financing throughout the Term of the ASAP III DBFM Project.

INFRA anticipates providing funding for the ASAP III DBFM Project in an amount equal to approximately 50% of the Successful Proponent's capital cost of the ASAP III DBFM Project. The Government of Alberta is currently considering the appropriate amount of this partial funding and the appropriate timing for payment of such partial funding. The final decisions on these points will be incorporated into the final draft of the DBFM Agreement issued to the Proponents prior to the RFP's SR Package 3 Submission deadline as set out in the ASAP III DBFM Project schedule in article 2.9, below.

Payment to the Successful Proponent will be based on the ongoing accessibility of the ASAP III DBFM Project schools and on achieving the specified performance requirements throughout the Term of the DBFM Agreement.

The monthly amount payable will be the amount stated in the Preferred Proponent's SR Package 3, as described in article 2.6, below. The Successful Proponents' monthly payments will be comprised of two components, a capital payment component and an M&R payment component.

The capital payment component for capital costs of the ASAP III DBFM Project schools will be made in fixed monthly amounts. The capital payment component will not be index linked to account for inflation. The M&R payment component will be index linked. All payments will be subject to payment adjustments should the ASAP III DBFM Project Schools become inaccessible (as will be defined in the DBFM Agreement) or should the Successful Proponent fail to achieve the specified performance requirements. Such payment adjustments will also be index linked. Additional one-time payments will be made for the work associated with the removal of modular classrooms or the addition of modular classrooms beyond those contemplated in the DBFM Agreement. The M&R payment component will also be adjusted for changes in the number of modular classrooms. Once removed, the Successful Proponent will no longer be responsible for the removed modular classroom and will no longer receive M&R payments for the removed modular classroom. Once added, the Successful Proponent will be responsible for the additional modular classroom to the same extent as for any existing modular classrooms and will receive M&R payments for the additional modular classroom.

2.6 Procurement Process

The first stage of the procurement process for the ASAP III DBFM Project is this RFQ. INFRA intends to invite three Respondents (but reserves the right to invite fewer or more than three Respondents) to participate, as Proponents, in the second stage of the procurement process, the RFP.

During the RFP stage, INFRA intends to require Proponents to submit the following packages in stages:

- Submission Requirement ("SR") Package 1 -
 - Proponent information including any changes to Consortium team membership (Refer to article 4.1); and
 - o general management and communication plans.
- SR Package 2
 - o detailed designs for the ASAP III DBFM Project schools;
 - detailed plans for design, construction (including but not limited to proposed construction schedule), commissioning, independent certification;
 - detailed plans for Maintenance, Renewal, site Maintenance/Renewal and condition monitoring and handback;
 - o quality management, environmental management, and LEED[™] certification plans; and
 - o plans for public communication, safety, collaboration and insurance.
- SR Package 2B –

Following SR Package 2 evaluation and clarification:

- a restated and revised SR Package 2 incorporating all of the clarifications provided by Proponents in response to INFRA's requests for clarification;
- o confirmation of the financial capacity of the Proponents and their Team Members;
- o indicative financial model;
- o indicative financing plan; and
- a letter from the Proponent stating that notwithstanding the contents of its SR Package 2B, the Proponent's Successful Proponent will design and construct the ASAP III DBFM Project in compliance with the Technical Requirements, as defined in the DBFM Agreement.

SR Package 2B is intended as the consolidated technical RFP submission from Proponents. Portions of SR Package 2B will be incorporated into the DBFM Agreement.

- SR Package 3
 - o final financial model, final financing plan and final financial offer.

SR Package 3 is intended as the consolidated financial RFP submission from Proponents. Portions of SR Package 3 will be incorporated into the DBFM Agreement.

As part of the RFP INFRA also intends to invite, but not require, Proponents to participate in an optional innovation submission and feedback process. The optional innovation submission ("**Optional Innovation Submission**") would include non-binding information regarding innovative solutions being considered and identify departures from the solutions presented in the Technical Requirements, as defined in the DBFM Agreement.

Proposed innovations that could require amendments to existing DP applications or could require altogether new DP applications to be made in relation to the ASAP III DBFM Project schools, thus creating schedule risk, will not be acceptable to INFRA.

If a Proponent does not intend to pursue its previously submitted innovation, that Proponent will not be required to adopt that innovation in its subsequent SR Packages.

Proponents will be afforded opportunities to engage in one-on-one technical and DBFM Agreement meetings with the ASAP III DBFM Project team.

The Proponent submitting a compliant proposal with the lowest net present value will be selected to proceed as the "**Preferred Proponent**".

An honorarium for RFP proposal development costs of \$750,000.00 (CDN) will be paid to each of the unsuccessful Proponents who have submitted a compliant proposal to the RFP and agrees to transfer to the Province all intellectual property rights (including waiving of moral rights) contained within the Proponent's proposal. The Successful Proponent will not be paid the honorarium.

2.7 Intentionally Deleted

2.8 DBFM Agreement and Risk Allocation

The allocation of risk between the public and private parties is fundamental to the success of the ASAP III DBFM Project. INFRA has developed a risk allocation for the ASAP III DBFM Project, based on the prior ASAP II Project, which in its view is likely to meet INFRA's objectives.

A draft DBFM Agreement will be issued concurrently with the RFP. It will be based, in large measure, upon the agreement used by INFRA in the ASAP II Project. A copy of this agreement is available on the Alberta Education website (see article 1.10).

During the RFP process Proponents will be afforded opportunities to provide written comments on the draft DBFM Agreement and thereafter engage in individual meetings with the ASAP III DBFM Project team. The DBFM Agreement will be finalized prior to the submission of SR Package 3 (see ASAP III DBFM Project Schedule article 2.9 immediately below), and **will not** be negotiated further following selection of the Preferred Proponent.

2.9 ASAP III DBFM Project Schedule

INFRA intends (but see the disclaimer in article 1.11, above) to adhere to the following schedule:

RFQ	
RFQ Response deadline (see article 1.4, above)	November 16, 2011
Respondent meetings with the RFQ selection committee	[TO BE DETERMINED; SEE ALSO
	ARTICLE 8.5]
RFP Proponent selection announcement	November 30, 2011
RFP (The deadline dates for interim draft DBFM Agreement comments by Proponents and notification dates of SR results by INFRA will be set out in the RFP.)	
Issue RFP	December 1, 2011
First DBFM Agreement meetings with Proponents	January 11-13, 2012
(individually)	
SR Package 1 Submission deadline	January 16, 2012
Optional Innovation Submission deadline	January 16, 2012
First technical meetings with Proponents (individually)	February 20-22, 2012
Second technical meetings with Proponents (individually)	March 14-16, 2012
Second DBFM Agreement meetings with Proponents	March 26-28, 2012
(individually)	
SR Package 2 Submission deadline	April 5, 2012
Issue final form of DBFM Agreement	May 18, 2012
Clarified SR Package "2B" Submission deadline	June 1, 2012
SR Package 3 Submission deadline	July 11, 2012
Notification of Preferred Proponent	July 24, 2012
DBFM Agreement signing (no later than)	September 24, 2012
	- ,

3 RFQ REQUIREMENTS

3.1 Ineligibility

Respondents must not include among their Team Members or Key Individuals any business entity or individual who through his/her employment or association (past or present) has had involvement with the ASAP III DBFM Project or projects related to the ASAP III DBFM Project including:

- any consultant retained by INFRA in relation to the ASAP III DBFM Project, including but not limited to consultants providing architectural, cost, technical performance specification writing, process, finance or financial capacity advice, namely:
 - ACI Architecture Inc.;
 - Al-Terra Engineering Ltd.;
 - BEI Engineering (2000) Inc.;
 - BPTEC-DNW Engineering Ltd.;
 - Douglas Walters Landscape Architect Ltd.;
 - Eclipse Geomatics & Engineering Ltd.;
 - Hukalo Oberg Engineering Ltd.;
 - KPMG LLP;
 - K.S.J. Engineering Ltd.;
 - Matrix Landscape Architecture Ltd.;
 - PricewaterhouseCoopers LLP;
 - Quinn Young Architects Ltd.;
 - Robertson & Associates Engineering Ltd.;
 - Tech-Cost Consultants Ltd.; and
 - TYZ Engineering Ltd.;
- R.B. Innes, the Fairness Auditor for the ASAP III DBFM Project;
- any member of the Legislative Assembly of Alberta, or any associated person (as set out in the *Conflicts of Interest Act* (Alberta));
- any former or current employee or representative of the ASAP III School Boards (as defined in article 1.1) who through his/her employment or association (past or present) has had involvement with the ASAP III DBFM Project or projects related to the ASAP III DBFM Project; or
- any former or current employee of the Province of Alberta, who through his/her employment (past or present) has had involvement with the ASAP III DBFM Project or projects related to the ASAP III DBFM Project;

or who is, or is associated with any party that is in any way likely to create a conflict of interest or a perception of conflict of interest.

If a Respondent considers that a particular relationship or association does not create a conflict of interest and will not create a perception of conflict of interest, but is concerned that INFRA could arrive at a different conclusion, the Respondent should fully disclose the circumstances to INFRA at the

earliest possible date, and request that INFRA provide an advance interpretation as to whether the relationship or association will be likely to create a conflict of interest or a perception of conflict of interest.

Failure to comply with this provision may result in disqualification of your Consortium from the RFQ process or, if INFRA becomes aware of your breach of this provision after the RFP has been issued, from the RFP process.

3.2 Team Name and Lead Contact

A Respondent's Response must at the outset state a team name and an organization (the "**Contact Organization**") that will be the contact for all communications with INFRA regarding this RFQ, and an individual (the "**Contact Individual**") responsible for all such communications on behalf of that organization. INFRA shall be entitled to rely on any communication from the Contact Individual as having been duly authorized by the Contact Organization and as being duly given on behalf of the Respondent and its Team Members.

The following information must be provided for the Contact Individual:

- name;
- company name;
- title;
- address;
- phone number;
- fax number; and
- e-mail address.

Table 3.2 in the attached Appendix "B" must be completed and provided as part of the Respondent's Response.

A Respondent may change its Contact Organization only by providing written notice from the previous Contact Organization. The Contact Organization may substitute a new Contact Individual only by a written notice to INFRA signed either by the previous Contact Individual or by an officer of the Contact Organization whose authority to do so is affirmed to the satisfaction of INFRA.

3.3 Confidentiality of Responses and the Collection of Personal Information

All records submitted in response to this RFQ may be subject to disclosure under the *Freedom of Information and Protection of Privacy Act* (Alberta) ("**FOIP**"). Except as otherwise expressly indicated by Respondents, Responses to this RFQ will be considered to have been submitted in confidence and will not be disclosed by INFRA or its consultants except as required by FOIP or any other law.

The Respondent must advise, and must ensure Team Members advise, Key Individuals that all personal information (as defined in FOIP) provided to INFRA in conjunction with the RFQ process is being collected for the purpose of evaluating the Respondent's Response, pursuant to the *Government Organization Act* (Alberta) and the terms of this RFQ. The Key Individuals must also be given the

name and contact information of the Designated Representative who may answer any questions they may have. It is the Respondent's responsibility to obtain Key Individuals' authorization to include such personal information in the Response and authorization for INFRA to do reference checks. Respondents must, if so requested by INFRA, supply evidence demonstrating that such authorizations have been properly obtained in accordance with this article.

Notwithstanding the foregoing, INFRA may release the identity of the Respondents, their Team Members and Key Individuals in relation to the ASAP III DBFM Project, as INFRA determines to be necessary.

3.4 Publicity

No press release shall be issued by any interested parties or Respondents or their Team Members or Key Individuals in relation to the ASAP III DBFM Project without first obtaining the prior written consent of INFRA. Breach of this provision may result in disqualification of your Consortium from the RFQ process or, if INFRA becomes aware of your breach of this provision after the RFP has been issued, from the RFP process.

3.5 No Collusion

Respondents must ensure that their participation in this RFQ is conducted without collusion or fraud on their part or any of their Team Members or Key Individuals. Respondents and their Team Members and Key Individuals shall not engage in discussions or other communications with any other Respondents or their Team Members and Key Individuals regarding the preparation or submission of their Responses. Breach of this provision may result in disqualification of your Consortium from the RFQ process or, if INFRA becomes aware of your breach of this provision after the RFP has been issued, from the RFP process.

3.6 No Common Ownership

Common ownership between Team Members on different Respondents ("**Common Ownership**") that may reduce, or be perceived to reduce, the competition between the applicable RFP proposals ("**Competition Reduction**") should such Respondents be short-listed may result in removal of the Respondent from the RFQ process. If a Respondent considers that a particular Common Ownership should not be determined to be Competition Reduction, but is concerned that INFRA could arrive at a different conclusion, the Respondent should fully disclose the circumstances to INFRA at the earliest possible date, and request that INFRA provide an advance interpretation as to whether the particular Common Ownership would be determined to be Competition Reduction. If INFRA determines a particular Common Ownership to be Competition Reduction, then INFRA may remove all but one of the Respondents involved in the Common Ownership (the "**Commonly-Owned Respondents**") from the RFQ process or, if the determination occurs after the RFP has been issued, from the RFP process.

INFRA may determine the Respondent involved in the Common Ownership that is not removed based upon the Respondent involved in the Common Ownership with the highest evaluation score pursuant to article 8 (RFQ Evaluation). If there is a tie among the highest evaluation scores among two or more Commonly-Owned Respondents, INFRA may determine the Respondent involved in the Common Ownership that is not removed based upon a random draw among the Commonly-Owned Respondents with the tied highest evaluation scores.

Whether INFRA determines that there is Competition Reduction is fact specific and depends upon a number of factors, including without limitation some of the following:

- the nature of the role the commonly-owned Team Members play with their respective Respondent, including without limitation whether the commonly-owned Team Members are Functional Lead Team Members with their respective Respondent;
- the size of the role the commonly-owned Team Members play in terms of the design, construction or M&R of the ASAP III DBFM Project;
- the anticipated public perception should the non-commonly-owned Proponent drop out of the RFP process for whatever reason leaving only the two Commonly-Owned Respondents. In such case, would the nature and degree of Common Ownership be such that the public may perceive competition to be lessened;
- whether the nature of the Common Ownership is indirect such as a Team Member's pension plan holds units in a fund and that fund holds shares in a Team Member with another Respondent;
- given similar facts, what have other jurisdictions in Canada and around the world done. Have other jurisdictions allowed the Commonly-Owned Respondents to be proponents under a request for proposals;
- would the Commonly-Owned Respondents be considered related, associated, or at non-arm's length under the *Income Tax Act* (Canada) and case law;
- any measures the Commonly-Owned Respondents propose to implement to give INFRA comfort that there are no communications (express or implied) between the commonly-owned Team Members in respect of the RFQ process or the RFP process;
- the number of Respondents; and
- whether the commonly-owned Team Members provide specialized services which can only commercially reasonably be provided by a very limited number of entities.

3.7 Non-Conforming Responses

If a Response is not strictly in accordance with any provision of this RFQ ("**Non-Conformance**"), INFRA may, at its option:

- if in INFRA's opinion the Non-Conformance is immaterial, waive the Non-Conformance;
- if the Non-Conformance is an omission, INFRA may give the Respondent up to five Business Days to supply the omitted material; or
- if in INFRA's opinion the Non-Conformance is material, reject the Response as non-compliant.

3.8 Expenses

No honorarium or compensation will be offered by INFRA to any Respondent, Team Member or Key Individual in consideration of expenses incurred in responding to this RFQ.

4 YOUR TEAM

4.1 Team Members

General Information Required of Team Members

Please list the Team Members who constitute your Consortium and their roles by completing Table 4.1A in the attached Appendix "B". You are required to present the members of the Consortium under the following four categories:

- 1. **Project Lead.** This consists of the Team Member or combination of Team Members that will direct and coordinate the activities of other Team Members. The Project Lead will be the directing mind and will of the Consortium;
- 2. **Design-Construction Team.** This consists of the Respondent's Team Members who will be responsible for the design and construction of the ASAP III DBFM Project;
- 3. **Maintenance & Renewal ("M&R") Team.** This consists of the Respondent's Team Members who will be responsible for the M&R of the ASAP III DBFM Project; and
- 4. **Financing Team.** This consists of the Respondent's Team Members who will be responsible for obtaining the financing for and financial management of the ASAP III DBFM Project.

For each of the three functional teams (namely Design-Construction, M&R, and Financing Teams), you must identify a lead member (the "**Functional Lead**") for each team. The Functional Lead Team Member shall be the Team Member responsible for the delivery of the work and the performance of the respective functional teams.

Respondents are strongly encouraged to provide résumés for each of the Key Individuals outlining each Key Individual's professional qualifications/designations, summary of education, and relevant experience.

Additionally, Respondents are strongly encouraged to provide an organizational chart(s) showing the organization of the functional teams and the identified Team Members (and details to show internal relationships between Team Members) as it relates to the entire ASAP III DBFM Project.

At this RFQ stage, it is mandatory that you name the Project Lead and the Functional Lead Team Members. It is not mandatory to name other Team Members that will make up the entire Consortium. You may name other Team Members in the RFQ Response. Where you anticipate that additional Team Members will be added to the Consortium following the short-listing of the Consortium as a Proponent, you should indicate in your response to article 6 ("Your Approach") how and when you intend to add such Team Members.

A particular Team Member may be part of more than one of those functional teams, described above, and thus be listed in more than one Table.

Information on each Team Member named is required in Tables 4.1B to 4.1I in the attached Appendix "B". Please fully complete these tables and include as part of your Response.

No Team Member is to be listed as part of your Consortium unless they have formally consented in writing to be so listed in your Response (which written consent must be produced if requested by INFRA).

Default and Litigation Disclosure of Team Members

Identify if any Team Member is currently in default on any loan agreement or financing agreement with any bank, financial institution, or other entity. If so, specify date(s), details, circumstances, and prospects for resolution.

Provide details of any current or pending litigation, or claims filed, against any of the Team Members.

Financial Information Required of Team Members

For <u>each</u> of the Functional Lead Team Members, as well as for any Team Member who is part of the Project Lead but is not otherwise a Functional Lead Team Member, and/or for any proposed guarantors and equity providers, please provide in a SEPARATELY SEALED ENVELOPED marked "CONFIDENTIAL FINANCIAL INFORMATION" the following information:

- Audited financial statements and annual reports for each of the last three years;
- Interim financial statements for each fiscal quarter since the most recent year for which audited statements are provided;
- Credit rating information, if available;
- Bank letters of reference (or, in the case of the Functional Lead Team Member of the Financing Team, such alternative information as in the Respondent's estimation will fully satisfy INFRA of the financial capability of such Functional Lead Team Member to lead and carry out your Consortium's plan for financing the ASAP III DBFM Project);
- Bonding capacity and letters of reference, if applicable;
- Details of any material events that may affect the entity's financial standing since the last annual or interim financial statement provided;
- Details of any bankruptcy, insolvency, *Companies' Creditors Arrangement Act* (Canada) arrangement or other insolvency litigation in the last three fiscal years; and
- Respondents are invited, but not required, to include a brief explanation of why they believe each Team Member has the financial standing, capacity and resources to carry out their respective roles on the ASAP III DBFM Project.

The above information may be replaced with equivalent financial information satisfactory to INFRA to demonstrate that the Project Lead, Functional Lead Team Members and any proposed or anticipated provider of equity who may not otherwise be a Project Lead or Functional Lead Team Member have sufficient financial standing, capacity, and resources to carry out their respective roles on the ASAP III DBFM Project.

If, at any time following the submission of a Response, a Consortium becomes aware of a change to the financial information required by this article; that Consortium must immediately so advise INFRA, and indicate its proposed response to the change.

INFRA reserves the right to reevaluate the Consortium's financial capacity to successfully secure or provide financing for, design and construct and perform M&R for the ASAP III DBFM Project having regard to the magnitude of the ASAP III DBFM Project.

4.2 Key Individuals

Tell us about the Key Individuals who will be the lead for each Team Member by completing, at a minimum, Tables 4.2/5.2A, 4.2/5.2B, 4.2/5.2C, and 4.2/5.2D in the attached Appendix "B". As indicated by those Tables, Key Individuals must be separately listed for:

- your Project Lead;
- your Design-Construction team;
- your M&R team; and
- your Financing team.

Providing resumes for Key Individuals as part of your Response to this RFQ is strongly encouraged. A particular Key Individual may be part of more than one of those teams, and thus be listed in more than one Table.

For each Key Individual, indicate your best estimate of the probability that the Key Individual will be available as needed throughout the ASAP III DBFM Project. Where appropriate, identify a proposed back-up or replacement for any Key Individual, and include the same information as if that back-up or replacement were a Key Individual.

4.3 Substitutions

If, following submission of your Response, you become aware that any of your Team Members or Key Individuals will be unable or is likely to be unable to participate in your Consortium, you must immediately so advise INFRA, and indicate your proposed substitute Team Member or Key Individual. Failure to do so may result in disqualification of your Consortium from the RFQ process or, if INFRA becomes aware of your breach of this provision after the RFP has been issued, from the RFP process.

Upon receipt of notification of a requested substitution of a Team Member or Key Individual, it is at the sole discretion of INFRA to allow such a substitution. In making a determination regarding the acceptability of a proposed substitution, INFRA, without limiting the generality of its considerations, will consider whether or not a proposed substitution:

- Provides an unfair advantage to the Respondent; or
- Jeopardizes or delays the RFQ evaluation process.

All information required by this RFQ in respect of Team Members and Key Individuals must be provided to INFRA in respect of any proposed substitute Team Members or Key Individuals, as applicable.

5 YOUR EXPERIENCE

5.1 Team Members' Project Experience

Tell us about the experience of each Team Member that is pertinent to the ASAP III DBFM Project by completing, at a minimum, Tables 5.1A, 5.1B, 5.1C, and 5.1D in the attached Appendix "B". As indicated by those Tables, the experience of Team Members must be separately listed for each of the following Teams:

- your Project Lead;
- your Design-Construction team;
- your M&R team; and
- your Financing team.

The Project Lead must indicate its past P3 experience and other relevant experience by completing Table 5.1A in the attached Appendix "B".

Where a Team Member is part of more than one of those teams, the pertinent experience of that Team Member must be listed in each applicable Table.

You should highlight your experience in schools and social infrastructure projects, as well as your experience in DBFM or other forms of private finance initiatives/public-private partnerships.

5.2 Key Individuals' Experience

Provide the experience of each Key Individual that is pertinent to the ASAP III DBFM Project by completing, at a minimum, Tables 4.2/5.2A, 4.2/5.2B, 4.2/5.2C, and 4.2/5.2D in the attached Appendix "B". As indicated by those Tables, the experience of Key Individuals must be separately listed for:

- your Project Lead;
- your Design-Construction team;
- your M&R team; and
- your Financing team.

Where a Key Individual is part of more than one of those teams, the pertinent experience of that Key Individual must be listed in each applicable Table.

You should highlight their experience in schools and social infrastructure projects, as well as their experience in DBFM or other forms of private finance initiatives/public-private partnerships. Each Key Individual should identify its experience on at least 3 previous projects.

6 YOUR APPROACH

6.1 Project Lead

Describe your approach for governing your Consortium through the RFP stage and during the design-construction phase of the ASAP III DBFM Project and beyond, having specific regard to:

- the role that the Project Lead will play in the organization, management and coordination of the Consortium including if applicable the Project Lead's role in managing the addition of Team Members to your Consortium, if your Consortium is short-listed as a Proponent;
- decision-making and issue resolution on behalf of the Consortium, both during the RFP stage and beyond;
- contractual relationships among Team Members;
- risk allocations among Team Members;
- the Project Lead's role, if any, in contributing equity and/or securing financing for the ASAP III DBFM Project;
- communications with media, the public, the applicable school boards, the school communities, the respective municipalities, and other stakeholders; and
- collaboration strategy for working with INFRA, Alberta Education and the school boards.

6.2 Design-Construction Team

Indicate the Functional Lead Design-Construction Team Member responsible for the overall design-construction. Additionally, provide any pertinent information regarding the intended organization, management and coordination of the Design-Construction Team.

Where the Design-Construction Team may not be fully formed, provide a clear plan that:

- demonstrates how the additional members of the team will be resourced;
- establishes the methodology for advertising, evaluating and selecting trade contractors and subcontractors; and
- describes your current relationships with the subcontracting community and which subcontractors may be well suited, in your opinion, to work on the ASAP III DBFM Project.

Describe the Design-Construction Team's approach for the design and construction of the ASAP III DBFM Project, having specific regard to:

- your approach to achieve LEED[™] Silver Certification from Canada Green Building Council for each school including modular classrooms;
- your understanding of challenges and opportunities associated with providing design and construction services for design-build projects;
- your approach for implementing industry's "best practices" for:
 - establishing and tracking project objectives;
 - o cost controls and tracking;
 - o change order management; and
 - o construction schedule control;

- your approach to quality management including:
 - o your quality assurance program with respect to design and construction;
 - the process by which you evaluate the life cycle costs of systems and materials;
 - the process by which the level of quality of systems and materials will be maintained; and
 - o total quality management for each phase, including commissioning;
- your approach to safety including:
 - o the process by which you will anticipate, recognize and manage safety risks;
 - the safety resources that the Design-Construction Team's Team Lead provides for each of its project's safety programs; and
 - compliance with Certificate of Recognition ("COR") requirements which are relevant to the applicable industry as issued by the Alberta Construction Safety Association or another certifying partner authorized to issue CORs by the Workplace Health and Safety and Employment Standards Compliance branch of Alberta Employment and Immigration or its successor branch or department of the provincial government of Alberta;
- your approach to the supply of modular classrooms;
- your approach to delivering the ASAP III DBFM Project schools on time including:
 - your approach to developing, maintaining and updating schedule during design and construction; and
- any specifically intended innovative approaches to design or construction.

6.3 *M&R Team*

Indicate the Functional Lead M&R Team Member responsible for the overall M&R. Additionally, provide any pertinent information regarding the intended organization, management and coordination of the M&R Team.

Where the M&R Team may not be fully formed, provide a clear plan that

- demonstrates how the additional members of the team will be resourced;
- establishes the methodology for advertising, evaluating and selecting trade contractors and subcontractors; and
- describes your current relationships with the subcontracting community and which subcontractors may be well suited, in your opinion, to work on the ASAP III DBFM Project.

Describe the M&R Team's approach for the M&R of the ASAP III DBFM Project, having specific regard to:

• your approach to Maintenance;

- your approach to Renewal;
- your approach to materials procurement;
- your approach to quality management including:
 - the process by which you evaluate the life cycle costs of systems and materials;
 - the process by which the level of quality of systems and materials will be maintained;
 - your quality control and quality assurance program with respect to changes in design and construction; and
 - your experience with the planning and coordination of energy conservation measures;
- your approach to environmental management;
- your approach to the supply (whether by procuring new or fixing existing modular classrooms supplied by INFRA) and addition or removal of modular classrooms to accommodate changing school enrolment;
- your approach to occupational health and safety and compliance with Certificate of Recognition ("COR") requirements which are relevant to the applicable industry as issued by the Alberta Construction Safety Association or another certifying partner authorized to issue CORs by the Workplace Health and Safety and Employment Standards Compliance branch of Alberta Employment and Immigration or its successor branch or department of the provincial government of Alberta;
- your approach to performance management, monitoring and reporting with an emphasis on identifying your processes for periodically measuring performance with respect to M&R;
- your approach to relationship management with school board representatives, the public and school user groups; and
- any specifically intended innovative approaches to M&R.

6.4 Financing Team

Indicate the Functional Lead Financing Team Member responsible for the overall financing. Each Consortium should outline its proposed approach to obtaining the required equity.

Additionally, provide any pertinent information regarding the intended organization, management and coordination of the financing team.

Where the Financing Team may not be fully formed, provide a clear plan that:

• demonstrates how the additional members of the team will be resourced;

- establishes the methodology for identifying, evaluating and selecting equity contributors or debt financing providers; and
- describes your current relationships with the financial community and which among them may be well suited, in your opinion, to contribute equity or provide financing for the ASAP III DBFM Project.

Describe your approach for financing of the ASAP III DBFM Project, having specific regard to:

- the delivery of security that will be required to be lodged upon execution of the DBFM Agreement (see articles 1.10 and 2.8);
- your anticipated financing structure and rationale;
- your anticipated financing sources for debt and equity;
- any potential financing issues you anticipate encountering during the RFP period, your experience with and your plan for dealing with such issues;
- any specifically contemplated alternatives to your financing plan;
- securing a financing package for the ASAP III DBFM Project Term that allows the Respondent, if selected to be a Proponent, to submit a Proposal on a fixed price basis;
- identifying perceived potential challenges to obtaining financing and proposed responses to satisfactorily address such challenges; and
- any specifically contemplated innovative approaches to financing the ASAP III DBFM Project.

7 YOUR ADVANTAGE

7.1 Why We Should Be Selected to Participate in the RFP Process

Under the heading "Why We Should Be Selected to Participate in the RFP Process", summarize in no more than five pages why the experience, organization, innovation or methodology of your Consortium gives you an advantage over other Respondents. In other words, why is your Consortium, if selected to participate in the RFP, more likely than other Respondents to:

- (a) engage in the procurement process through to final bids;
- (b) submit a highly competitive final bid that would achieve value for money for the ASAP III DBFM Project;
- (c) secure a competitive financing package in a timely manner; and
- (d) if selected as the Successful Proponent, design, build and maintain the schools safely, effectively and efficiently, in a manner that is financially and environmentally sound, and that appropriately protects the health and safety of the students and other school users.

In addition to what is required above, Respondents may wish to highlight any prior collective experience of Team Members or Key Individuals participating together on similar projects in their Responses. Outlining such prior collective experience through the completion of Table 7.1 is strongly encouraged (and Table 7.1 is not included for the purpose of the five page limit referenced above).

8 **RFQ EVALUATION**

8.1 Completeness and Compliance Check

Responses will be checked for completeness and compliance prior to more detailed evaluation according to the Scoring Matrix as described in the article 8.2. The completeness and compliance check will be based on the following:

- Receipt of the complete Response on or before the time as set out in article 1.4, above;
- Complete information on the Contact Organization and Contact Individual according to Table 3.2;
- Completion of all required Tables [Tables 3.2 to Tables 5.1D]; and
- Submission of a separately sealed envelope containing "Confidential Financial Information" (see articles 1.5 and 4.1).

Responses that do not meet the above criteria may be rejected and may not be considered any further.

8.2 Scoring Matrix

Subject to article 8.3, below, INFRA intends to evaluate Responses in accordance with the following scoring matrix:

Evaluation Category	Weighting (%)
Project Lead	25
Organization, Role and Plan	10
Team Members' Experience	10
Key Individuals' Experience	5
Design - Construction Team	25
Organization. Approach and Plan	10
Team Members' Experience	5
Design Key Individuals' Experience	5
Construction Key Individuals' Experience	5
Maintenance and Renewal Team	20
Organization, Approach and Plan	10
Team Members' Experience	5
Key Individuals' Experience	5
Financing Team	25
Organization, Approach and Plan	15
Team Members' Experience	5
Key Individuals' Experience	5
Your Advantage	5
Total	100

8.3 Risk Assessment of Respondents

In addition to the scoring matrix in article 8.2, Respondents will be evaluated on a pass-fail basis based on INFRA's assessment of:

- financial capacity to successfully design and construct the ASAP III DBFM Project; and
- capability of arranging the financing required for the ASAP III DBFM Project;

in each case having regard to the magnitude of the ASAP III DBFM Project.

8.4 Basis of Evaluation

Respondents will be evaluated primarily on the basis of their Responses, including any additional clarifications or information supplied pursuant to requests from INFRA under article 8.5, below. In addition, INFRA may consider information received from any source that INFRA considers reliable, including but not limited to:

- research to validate information supplied by Respondents;
- information and opinions supplied by INFRA's consultants; and
- banking and reference checks (to which banking and reference checks Respondents and their Team Members and their Key Individuals shall be deemed to have consented by the Respondent delivering its Response to INFRA).

8.5 Clarifications and Interviews

If requested by INFRA to provide clarification or additional information in relation to a Response to this RFQ, a Respondent must provide such clarification or additional information within the time specified by INFRA.

If requested by INFRA a Respondent must, at the time and location specified by INFRA, meet with representatives of the ASAP III DBFM Project team or the RFQ selection committee to provide explanation or clarification of its Response to this RFQ (see article 2.9 ASAP III DBFM Project Schedule).

8.6 Debriefing

Following completion of the evaluation process and announcement of the Respondents short-listed as Proponents for participation in the RFP, INFRA will upon request conduct an individual debriefing session with any unsuccessful Respondent.

During such debriefing, the evaluation, scoring, ranking and content of any Responses will not be disclosed. Only the strengths and weaknesses of the Respondent's Response relative to the evaluation criteria will be disclosed and discussed.

APPENDIX "A" (article 1.12 – Defined Terms)

"**ASAP III DBFM Project**" means the thirteen new schools in central and southern Alberta as described in article 2.1;

"Business Day" means a day other than a Saturday, Sunday or statutory holiday in Alberta;

"Consortium" means a team of private sector participants;

"Contact Individual" has the meaning as set out in article 3.2;

"Contact Organization" has the meaning as set out in article 3.2;

"**COR**" means a Certificate of Recognition recognized by the Workplace Health and Safety and Employment Standards Compliance branch of Alberta Employment and Immigration;

"DBFM" means design, build, finance and Maintenance and Renewal;

"Designated Representative" means the individual on behalf of INFRA set out in article 1.7;

"FOIP" means Freedom of Information and Protection of Privacy Act (Alberta);

"Functional Lead" has the meaning as set out in article 4.1;

"**INFRA**" means Her Majesty the Queen in right of Alberta, as represented by the Minister of Infrastructure;

"**Key Individuals**" means the individuals that will play a lead role on behalf of a particular Team Member;

"Maintenance" means the upkeep, repair and replacement of all building components of all the schools of the ASAP III DBFM Project;

"Maintenance and Renewal" or **"M&R"** means collectively the upkeep, repair and replacement of all building components and the life-cycle replacement of major building components and site rehabilitation of all the schools of the ASAP III DBFM Project;

"Optional Innovation Submission" as described in article 2.6;

"**Preferred Proponent**" means the Proponent, which may be selected by INFRA, submitting a compliant bid to the RFP whose bid is a payment stream having the lowest net present value over the Term as determined by INFRA;

"Proponents" means the Respondents short-listed to participate in the RFP;

"Registered Parties" means interested parties registered with INFRA pursuant to article 1.4;

"**Renewal**" means the life-cycle replacement of major building components and site rehabilitation of all the schools of the ASAP III DBFM Project;

"Respondents" means Consortia responding to the RFQ;

"Response" means all of the information submitted by a Respondent in response to the RFQ;

"**RFP**" means Request for Proposals;

"RFQ" means this Request for Qualifications;

"Submission Requirement" or "SR" means those requirements specified in the ASAP III DBFM Project RFP;

"**Successful Proponent**" means the legal entity, as selected by the Preferred Proponent, and as approved in advance and in writing by the Province acting reasonably, that will enter into the DBFM Agreement;

"Team Members" means the key business entities forming a particular Respondent;

"Term" means the design-construction phase and the 30-year period thereafter.

APPENDIX "B"

- 1. Table 3.2: Contact Organization and Contact Individual
- 2. Table 4.1A: Team Members
- 3. Table 4.1B: Project Lead Team Member Name and Role
- 4. Table 4.1C: Project Lead Team Member Legal Status
- 5. Table 4.1D: Design-Construction Team Members Name and Role
- 6. Table 4.1E: Design-Construction Team Member Legal Status
- 7. Table 4.1F: M&R Team Name and Role
- 8. Table 4.1G: M&R Team Member Legal Status
- 9. Table 4.1H: Financing Team Member Name and Role
- 10. Table 4.11: Financing Team Member Legal Status
- 11. Table 4.2 / 5.2A: Project Lead Team Key Individuals
- 12. Table 4.2 / 5.2B: Design-Construction Team Key Individuals
- 13. Table 4.2 / 5.2C: M&R Team Key Individuals
- 14. Table 4.2 / 5.2D: Financing Team Key Individuals
- 15. Table 5.1A: Project Lead Team Member P3 or Other Relevant Project Experience
- 16. Table 5.1B: Design-Construction Team Member Experience
- 17. Table 5.1C: M&R Team Member Experience
- 18. Table 5.1D: Financing Team Member Experience
- 19. Table 7.1: Your Advantage Collective Experience

Table 3.2 - Contact Organization and Contact Individual

Name of Dagnon dant	
Name of Respondent Contact Organization Mailing Address	Address: City: Province/State: Country: Postal Code:
Contact Organization Telephone Number	
Contact Organization Fax Number	
Contact Organization E-mail Address	
Contact Organization Web-site Address	
Name of Contact Individual	
Title	
Company	
Mailing Address	Address: City: Province/State: Country: Postal Code:
Telephone Number	
Fax Number	
E-mail Address	

Table 4.1A - Team Members

Team Member Category	Team Member (Company/Firm Name)	Indicate if Functional Lead	Primary Role and Responsibility	Key Individuals with the Team Member (Name, Title and brief description of primary role and responsibility) [NOTE: each Key Individual listed below must also be listed in Tables 4.2/5.2A through 4.2/5.2D in which more detail may be provided]
Project Lead		N/A		
•		N/A		
•		N/A		
Design-Construction				
•				
•				
M&R				
•				
•				
Financing				
•				
•				

<u> Table 4.1B – Project Lead Team Member Name and Role</u>
[NOTE: repeat this table for each Team Member that constitutes the Project Lead]

Name	
Mailing Address	Address:
	City:
	-
	Province/State:
	Country:
	Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Web-site Address	
Key Individual that is	
the contact person for the	
Team Member	
	Address:
Mailing Address	City:
	Province/State:
	Country:
	Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Role and Responsibility of	
Team Member in Project	
Lead Team	

Table 4.1C – Project Lead Team Member Legal Status

Project Lead Team Member:	
Type of Entity (corporation, partnership, joint venture, etc.)	
Legal Name	
Jurisdiction of Incorporation / Registration	
Registration No.	
Year of Incorporation / Registration	
Registered Address	
Current Trading/Business Name	
For Privately Held Corporation provide Director list	
Provide a description setting out (or diagram showing) the	
legal / beneficial ownership chain of the Team Member from	
the direct owner(s). If a particular level of ownership chain	
has diffuse ownership among numerous persons, that level	
can be summarized in the description/diagram.	

Table 4.1D	– D)es	sig	n-C	<u>'onst</u>	ru	ction	Te	am	N	ſen	nbe	er N	lam	e ai	nd I	<u>Role</u>
					-			-			_		~				-

	[NOTE: repeat this table for each Team Member in the Design-Construction Team]
Name	
Mailing Address	Address: City: Province/State: Country: Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Web-site Address	
Key Individual that is the contact person for the Team Member	
Mailing Address	Address: City: Province/State: Country: Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Role and Responsibility of Team Member in Design- Construction Team	

[NOTE, report this table for each Team Member in the Design Construction Team]

<u>Table 4.1E – Design-Construction Team Member Legal Status</u> [NOTE: repeat this table for each Team Member in the Design-Construction Team]

Design-Construction Team Member

Type of Entity (corporation, partnership, joint venture, etc.)	
Legal Name	
Jurisdiction of Incorporation / Registration	
Registration No.	
Year of Incorporation / Registration	
Registered Address	
Current Trading/Business Name	
For Privately Held Corporation provide	
Director list	
Provide a description setting out (or	
diagram showing) the legal / beneficial	
ownership chain of the Team Member from	
the direct owner(s). If a particular level of	
ownership chain has diffuse ownership	
among numerous persons, that level can be	
summarized in the description/diagram.	

Table 4.1F – M&R Team Name and Role [NOTE: repeat this table for each Team Member in the M&R Team]

	[10112. repeat this table for eac	n Team Member in the M&K Team]
Name		
Mailing Address	Address: City: Province/State: Country: Postal Code:	
Telephone Number		
Fax Number		
E-mail Address		
Web-site Address		
Key Individual that is the contact person for the Team Member		
Mailing Address	Address: City: Province/State: Country: Postal Code:	
Telephone Number		
Fax Number		
E-mail Address		
Role and Responsibility of Team Member in M&R Team		

Table 4.1G - M&R Team Member Legal Status [NOTE: repeat this table for each Team Member in the M&R Team]

M&R Team Member:	
Type of Entity (corporation, partnership,	
joint venture, etc.)	
Legal Name	
Jurisdiction of Incorporation / Registration	
Registration No.	
Year of Incorporation / Registration	
Registered Address	
Current Trading/Business Name	
For Privately Held Corporation provide	
Director list	
Provide a description setting out (or	
diagram showing) the legal / beneficial	
ownership chain of the Team Member from	
the direct owner(s). If a particular level of	
ownership chain has diffuse ownership	
among numerous persons, that level can be	
summarized in the description/diagram.	

			_			_	_	_	_	_	_			_	_		
able	4.1H	_	F	'i ı	nar	ncing	Ί	leam	Ν	Лe	mb	er	N	ame	and	l Rol	e
OTT			•			0		T							•	T	

Table 4.1H – Financing Team Member Name and Role
[NOTE: repeat this table for each Team Member in the Financing Team]

	[NO1E: repeat this table for each Team Member in the Financing Team]
Name	
Mailing Address	Address:
-	City:
	Province/State:
	Country:
	Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Web-site Address	
Key Individual that is	
the contact person for the Team	
Member	
Mailing Address	Address:
	City:
	Province/State:
	Country:
	Postal Code:
Telephone Number	
Fax Number	
E-mail Address	
Role and Responsibility of	
Team Member in Financing	
Team	

<u>Table 4.11 – Financing Team Member Legal Status</u> [NOTE: repeat this table for each Team Member in the Financing Team]

Financing Team Member:	
Type of Entity (corporation, partnership, joint venture, etc.)	
Legal Name	
Jurisdiction of Incorporation / Registration	
Registration No.	
Year of Incorporation / Registration	
Registered Address	
Current Trading/Business Name	
For Privately Held Corporation provide Director list	
Provide a description setting out (or diagram showing) the legal / beneficial ownership chain of the Team Member from the direct owner(s). If a particular level of ownership chain has diffuse ownership among numerous persons, that level can be summarized in the	
description/diagram.	

<u> Table 4.2/5.2A – Project Lead Key Individuals</u>

Key Individual Name [NOTE: Key Individuals listed below must also be listed in Table 4.1A]	Employing Team Member	Key Individual Role in Project	Key Individual Years of Experience	Selected Past Project Name and Description	Role of Key Individual on Past Project	Past Project Total Capital Cost and Term of Project in Years	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)

Selected Past Key Employing Key Key **Past Project Client Reference Role of Key** Individual **Design-**Individual Individual **Project Name and Total Capital** (Client Name, Contact Individual Name Construction Role in Years of Description on Past **Cost and Term** name, Phone number, Project Team Project Related of Project in Fax, E-mail) [NOTE: Key **Please indicate** Individuals Member Experience Years nature of project listed below must also be (e.g. P3, designlisted in Table build, design-bid-4.1A] build)

 Table 4.2/5.2B – Design-Construction Team Key Individuals

Table 4.2/5.2C – M&R Team Key Individuals

Key Individual Name [NOTE: Key Individuals listed below must also be listed in Table 4.1A]	Employing M&R Team Member	Key Individual Role in Project	Key Individual Years of Related Experience	Selected Past Project Name and Description Please indicate nature of project (e.g. P3, property management)	Role of Key Individual on Past Project	Past Project Average Annual M&R Value and Term of M&R Project in Years	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)

<u> Table 4.2/5.2D – Financing Team Key Individuals</u>

Key Individual [NOTE: Key Individuals listed below must also be listed in Table 4.1A]	Employing Financing Team Member	Key Individual Role in Project	Key Individual Years of Related Experience	Selected Past Project Name and Description	Type and Amount of Financing Raised (including capital structure, sources of financing and key features of financing)	Role of Key Individual on Past Project	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)

<u>Table 5.1A – Project Lead Team Member P3 or Other Relevant Project Experience</u> [NOTE: repeat this table for each Team Member that constitutes the Project Lead identified in Table 4.1A]

Project Lead Team Men	1ber:			
Past P3 or Relevant	Client Reference	Project Lead Team	Past Project Total	Date of Financial Close
Project	(Client Name, Contact	Member Role on Past	Capital Cost (whether	and Construction
Name/Description	name, Phone number,	P3 or Other Relevant	P3 or Relevant Project)	Completion for Past P3
	Fax, E-mail)	Project	and Term of Project in	or Relevant Project
			Years	
P3 Projects with				
financial close within				
past two years:				
Other P3 Projects:				
Other Relevant Projects				

<u>Table 5.1B – Design-Construction Team Member Experience</u> [NOTE: repeat this table for each Team Member in the Design-Construction Team identified in Table 4.1A]

Design-Construction Team Member:

Past Project Name/Description Please indicate nature of project (e.g. P3, design-build, design- bid-build)	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)	Design-Construction Team Member Role on Past Project	Past Project Total Capital Cost and Term of Project in Years	Was Team Member the Design-Construction Lead of the Past Project?
Projects within past two years:				
Other projects:				

<u> Table 5.1C – M&R Team Member Experience</u>

[NOTE: repeat this table for each Team Member in the M&R Team identified in Table 4.1A]

M&R Team Member:

Past Project Name / Description Please indicate nature of project (e.g. P3, property management)	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)	M&R Team Member Role on Past Project	Past Project Average Annual M&R Value and Term of M&R Project in Years	Was Team Member the M&R Lead of the Past Project?
Projects within past two years:				
Other projects:				

Table 5.1D – Financing Team Member Experience

[NOTE: repeat this table for each Team Member in the Financing Team identified in Table 4.1A]

Financing Team	Member:		1	1	
Past Project Name and Description	Client Reference (Client Name, Contact name, Phone number, Fax, E-mail)	Type and Amount of Financing Raised (include capital structure, sources of financing and key features of financing.)	Date of Financial Close	Role of Team Member on Past Project	Was Team Member the Financing Lead of the Past Project?
Projects within					
past two years:					
Other projects:					

Project Name/Description /Stage of Procurement	Team Member 1 (NOTE: Team Members listed below must also be listed in Table 4.1A)	Team Member 2 (NOTE: Team Members listed below must also be listed in Table 4.1A)	Etc.	Key Individual 1 (NOTE: Key Individuals listed below must also be listed in Table 4.1A)	Key Individual 2 (NOTE: Key Individuals listed below must also be listed in Table 4.1A)	Etc.	
Project A	•	•		•			Bullet indicates Team Member/Key Individual worked on Project
Project B		•	•		•		Bullet indicates Team Member/Key Individual worked on Project
Project C	•	•	•				Bullet indicates Team Member/Key Individual worked on Project
Etc.							Bullet indicates Team Member/Key Individual worked on Project

Table 7.1:	Your	Advantage	- Collective	Experience
1 and / .1.	Tour	Auvantage	- Concentre	Experience

Exhibit "A" (article 1.1 and 2.1– school sites, Page 1 of 1)

